Author’s response to reviews

Title: Study Protocol. The reduction of intoxication and disorder in premises licensed for the serving of alcohol: an exploratory randomised controlled trial

Authors:

Simon C Moore (mooresc2@cardiff.ac.uk)
Brennan Iain (l.Brennan@hull.ac.uk)
Murphy Simon (MurphyS7@cf.ac.uk)
Ellie Byrne (byrnee@Cardiff.ac.uk)
Jonathan P Shepherd (shepherdjp@Cardiff.ac.uk)
Moore Laurence (moorel1@cf.ac.uk)
Moore Susan (MooreS4@cardiff.ac.uk)

Version: 3 Date: 8 October 2010

Author’s response to reviews:

Thank you for allowing us to respond to queries concerning our manuscript "The reduction of intoxication and disorder in premises licensed for the serving of alcohol: an exploratory randomised controlled trial".

A Competing Interests section has been appended as a section detailing authors' contribution. With regards the latter, you asked that we identify authors by their initials. As you will see, three authors share the same initials and for reasons of clarity we have included names in this section.

Finally, we have elaborated on the study conclusions in the Abstract.